Author's response to reviews

**Title:** Reduced uptake of the proliferation-seeking radiotracer 99mTc-(V)DMSA in a 47-year-old woman with severe breast epithelial hyperplasia taking ibuprofen: a case report

**Authors:**

Vassilios J Papantoniou (vpapantoniou@gmail.com)
Evangelia K Sotiropoulou (evasot@gmail.com)
Pipitsa N Valsamaki (pivals@gmail.com)
Angeliki G Tsaroucha (angelikitsaroucha@hotmail.com)
Maria G Sotiropoulou (marsot2@yahoo.com)
Nikolaos D Ptohis (nikptohis@yahoo.gr)
Aikaterini J Stipsanelli (cgbats@gmail.com)
Konstantinos E Dimitrakakis (dimitrac@mail.nih.gov)
Spyridon G Marinopoulos (spyro@hol.gr)
Spyridon T Tsiouris (spytsi@gmail.com)
Aris J Antsaklis (arisants@otenet.gr)

**Version:** 3  **Date:** 28 January 2010

**Author's response to reviews:** see over
Athens, 28th January, 2010

To the Editor In Chief

Dear Sir/Madam,

We have the honour to resubmit the final version of our accepted manuscript entitled: “Reduced uptake of the proliferation-seeking radiotracer $^{99m}$Tc-(V)DMSA in a 47-year-old woman with severe breast epithelial hyperplasia taking ibuprofen: a case report”.

We proceeded with the major revisions (including ensuring that the author names are identically in the manuscript and the submission system and that the ethnic origin and nationality of the patient are provided in the manuscript) as well as the minor revisions (including typographical errors and final revisions). We remain at your disposal for any matter that may arise

Thank you in advance

Kindest regards

Vassilios Papantoniou, MD, PhD
Head of the Nuclear Medicine Department